

DANGEROUS DRUGS ACT

ANNUAL REPORT 2015 and 2016



Table of Contents

Introduction	4
Background	4
The Dangerous Drugs Act	4
2. National Responsibilities	6
Performance of Activities	7
Part IIA – Offences relating to Precursor Chemicals	9
Part III – Supply of Dangerous Drugs by Pharmacists under Licence and Appointm	nent of
Part IV: Summary Convictions	10
National Responsibilities	10
Part VIIA: Offences on the High Seas	10
Part IX: Miscellaneous	11
± ±	
Appendix II: Reports Submitted to the International Narcotics Control Board (INC	B) in 2015.
Appendix III: Reports Submitted to the International Narcotics Control Board (INC	CB) in 2016.
Appendix VII: Licences to Sell and Distribute Narcotics Issued in 2016	36

Introduction

This report is laid in the both Houses of Parliament as required pursuant to section 59A of the Dangerous Drugs Act Chapter 11:25 as amended.

Background

The Government and people of Trinidad and Tobago recognise that production, trafficking, and consumption of illegal drugs are disruptive to public health, safety, and the social fabric.

Passage of the Dangerous Drugs Amendments in 1994 brought the laws of Trinidad and Tobago in conformity with the requirements of the 1988 United Nations Convention. These amendments cover precursor chemicals, money laundering, maritime trafficking, extradition of money launders, forfeiture of assets and removal of impediments to effective prosecution.

The Dangerous Drugs Act Chapter 11:25 is an Act to provide for the control for narcotic drugs and psychotropic substances, and to make provision for the confiscation of the proceeds of drug trafficking and other provisions in connection with drug trafficking and matters connected therewith. Parts V to V111, containing sections 30 to 53, were repealed by the Proceeds of Crime Act number 55 of 2000. Notwithstanding any proceedings in respect of drug trafficking offences which had commenced before the proclamation of this Act proceeded as if sections 30 to 53 of the Dangerous Drugs Act continue to have effect. Act 55 of 2000 was proclaimed on the 6th day of November 2000 by Legal Notice 269/2000.

The Dangerous Drugs Act

This Act is administered by the Ministers of Health, National Security, and Foreign Affairs. Notwithstanding the Minister of Health pursuant to section 59A, is mandated to cause to be laid in both Houses of Parliament, annually, a report on the administration of the Act.

- Part I addresses the preliminary provisions which include the definition of terms/words used in the Act. It also provides that it shall have effect even though it is inconsistent with sections 4 and 5 of the Constitution.
- Part II addresses the possession of and trafficking in dangerous drugs. It also empowers the Minister of Health to make Regulations for inter alia the issue of licences for the import, export, diversion, sale, manufacture, production, or distribution, at a stated place, of any dangerous drug; and issue licences for the cultivation, gathering, or production, at a stated place, of opium poppy, marijuana, or coca plant. Further it provides for a number of actions to be offences, and prescribes the punishment where a conviction occurs.
- Part II A deals with offences relating to Precursor Chemicals. It also gives the Minister of
 Health the power to amend the Fourth Schedule by addition or deletion therefrom any
 substance, the inclusion or exclusion of which is by him deemed necessary in the public

- interest. It also empowers the Minister to make Regulations such as for the keeping of records and the labelling of consignments of scheduled substances.
- Part III addresses the supply of dangerous drugs by Pharmacist under licence and the appointment of Inspectors and Analysts. It also contains a number of offences and the penalties attached thereto.
- Part IV- addresses summary convictions.
- Part IV A provides for the Burden of Proof and Presumptuous.
- Part V111A addresses Offences on the High Seas.
- Part IX contains miscellaneous provisions such as the power of the Ministers of Health and Finance to make Regulations. It also repeals the Narcotic Control Ordinance and the Narcotic Drugs and Psychotropic Substances Control Act.
- The first schedule contains a list of Narcotic Drugs,
- The second schedule contains a list of Psychotropic Substances,
- The third schedule contains amendments to the Interpretation Act,
- The fourth schedule contain a list of Precursor Chemicals, and
- The fifth schedule addresses enforcement powers in respect of ships.

Roles and Responsibilities of the Drug Inspectorate with respect to the Enforcement of the Dangerous Drugs Act No. 38 of 1991 and its Amendment Act 27 of 1994

Currently the Drug Inspectorate controls and monitors narcotics and preparations containing narcotics.

1. International Responsibilities

1.1. United Nations Single Convention on Narcotic Drugs 1961

Activities

- Quarterly Statistics of Import and Exports of Narcotic drugs Form A
- Annual Estimates of Requirements for Narcotic Drugs, Manufacture of Synthetic Drugs, and Cultivation of the Opium Poppy, the Cannabis Plant and the Coca Bush – Form B
- Annual Statistics of Production, Manufacture, Consumption, Stocks and Seizures of Narcotic Drugs – Form C
- 1.2. United Nations Convention on Psychotropic Substances 1971

Activities

- Annual Statistical Report on Substances Listed in the Convention on Psychotropic Substances of 1971 – Form P
- 1.3. United Nations Convention against Illicit traffic in Narcotic drugs and Psychotropic Substances, 1988

Activities

• Preparation of "Annual Information on Substances Frequently used in the Illicit Manufacture of Narcotic Drugs and Psychotropic Substances" – **Form D**

1.4. Comisión Interamericana para el Control del Abuso de Drogas (CICAD)

(Inter-American Drug Abuse Control Commission)

• Preparation of the Multilateral Evaluation Mechanism (MEM) indicators for CICAD

2. National Responsibilities

- 2.1.1. Monitoring and controlling narcotics in accordance with relevant legislation by verification and processing of relevant applications for annual narcotic licences to pharmacies, pharmaceutical firms and Licences, Certificates of Official Approval to Import and Approval of Export for Narcotics.
- 2.1.2. Visiting pharmacies, pharmaceutical firms, nursing homes and private hospitals, to inspect premises, stocks and records of narcotics in compliance with relevant legislation.
- 2.1.3. Inspection, verification and destruction of expired narcotics.
- 2.1.4. Inspection of all new and relocated premises to ensure compliance with relevant legislation and institutional guidelines.

- 2.1.5. Audit and take-over of narcotic stocks and records of closure of antibiotic and narcotic cupboards when licensed pharmacist is leaving and there is no incoming pharmacist.
- 2.1.6. Hand-over of narcotic stocks and records and reopening narcotic cupboards when there is an incoming pharmacist.
- 2.1.7. Investigation of any breaches of narcotic legislation and taking of necessary action.
- 2.1.8. Maintenance of Narcotic Inspection and Application files.
- 2.1.9. Monitoring the movement of narcotics into and out of the country. This is facilitated through the receipt and audit of transaction reports from all importers monthly and maintenance of database for narcotic sales.
- 2.1.10. Destruction of Police Narcotic exhibits.
- 2.1.11. Verification and endorsement of "Export Declarations for Narcotics" from "Competent Authorities" (Regulatory Authority of a foreign country).
- 2.1.12. Transmission of copies of Certificates of Official Approval to Import Narcotics to "Competent Authorities".
- 2.1.13. Transmission of Copies of Certificates of Official Approval to Import Narcotics to Customs & Excise Division.
- 2.1.14. Registration of new Narcotic Preparations and Preparations containing narcotics for use by the population.

Performance of Activities

All of our international responsibilities have been met and completed forms have been submitted, in most cases, within the stipulated deadlines.

Notwithstanding a critical staff shortage, all of our national responsibilities with respect to the monitoring and controlling of narcotics have been met; with the exception of items 2.1.2 and 2.1.10 above.

Despite the absence of regulations and our critical staff shortage, most of our responsibilities with respect to controlling and monitoring of precursor chemicals have been met. However, activities such as visiting factories and chemical distributors to inspect stocks, usage and records, monitoring the movement of precursor chemicals out of the country, and maintaining a database on precursor chemicals have not been done.

Statistical reports for the years 2015 and 2016 can be found in Appendices I, II, and III.

Dangerous Drugs Act No. 38 of 1991

Part II - Possession of and Trafficking in Dangerous Drugs

- 4. The Minister may, subject to Regulations made under section 57
 - (a) issue licences for the import, export, diversion, sale, manufacture, production, or distribution, at a stated place, of any dangerous drug;

For the year 2015 (see Appendix IV):

- 100 applications to import narcotics were received and processed;
- 96 applications were approved; 4 applications were not approved;
- 96 licences were issued;
 - 14 of the issued licences were subsequently cancelled at the request of the licensee.

For the year 2016 (see Appendix V):

- 96 applications to import narcotics were received and processed;
- 95 applications were approved; 1 application was not approved;
- 95 licences were issued; and
 - 7 of the issued licences were subsequently cancelled at the request of the licensee.
- (b) issue licences for the cultivation, gathering or production, at a state place, of opium poppy, marijuana, or coca plant;
 - No applications were received and no licences were issued in 2015 or 2016.
- (c) name the ports or places in the Territory where any dangerous drug may be exported or imported;
- (d) prescribe the manner in which any dangerous drug is to be packed and marked for export;
- (e) authorise the furnishing of dangerous drugs to the master of a ship for the medicinal needs of the crew thereof;
- (f) prescribe the records that are to be kept by any person in connection with the export, import, manufacture, receipt, sale, disposal, and distribution of a dangerous drug.
- 5. Subsection 2, Section 5 (1) provides that any person who is in possession of a dangerous drug is guilty of an offence and is liable to a fine or imprisonment if convicted. Section 5 (3) A provides that any persons who cultivates gathers or produces any marijuana, except where he does so under a licence or acting under the supervision of a licence holder commits an offence and if convicted is liable to a fine or imprisonment. Section 5 (4) addresses the issue of the trafficking of a

dangerous drug, which is also an offence, and if convicted the person is liable to fine or imprisonment.

- 6. The Court before which a person is convicted under section 5, may, before imposing a sentence, order the Psychiatric Hospital Director to admit the person convicted to the psychiatric hospital named in the Order.
 - No persons convicted under Section 5 were sent to St Ann's Hospital pursuant to Section 6 for 2015 or 2016.
- (7A) Notwithstanding subsections (3A), (5), (6) and (7), where a person is charged for a drug trafficking offence under this Act, the Director of Public Prosecutions may, at the preliminary enquiry, elect to proceed with the matter summarily and if the accused so consents, the Court may adjourn the matter to be dealt with accordingly.
 - No information is currently available.

Part IIA - Offences relating to Precursor Chemicals

- 6B. The Minister by Order, add substances to the Fourth Schedule and may delete therefrom any substance, the inclusion or exclusion of which is by him deemed necessary in the public interest.
 - No additions or deletions of any substances to the Fourth Schedule were made in 2015 or 2016.
- 6C. (1) The Minister may make Regulations
 - (a) for imposing requirements as to the documentation of transactions involving substances referred to in the Fourth Schedule
 - No new Regulations were made in 2015 or 2016.

Part III - Supply of Dangerous Drugs by Pharmacists under Licence and Appointment of Inspectors and Analysts

- 7. (1) A person who has a licence under section 4 to deal in any dangerous drug shall not supply it to any person except as specifically provided in this Act.
 - 445 licences to sell and distribute narcotics were issued in 2015 (see Appendix V).
 - 452 licences to sell and distribute narcotics were issued in 2016 (see Appendix VI).
- (2) A person who has a licence under section 4 to deal in any dangerous drug may, upon receiving a written order therefor dated and signed by a person who has a licence under section 4 to deal in a dangerous drug, or by a pharmacist, medical practitioner, dentist or veterinary surgeon, whose signature is known to the person receiving the order, or if unknown, then verified before

the order is filled, supply a dangerous drug to such person, or to a person authorized under the Regulations to purchase or be in possession of a dangerous drug.

- 10. A person who contravenes section 7, 8 or 9 commits an offence and is liable upon conviction on indictment to a fine of one hundred thousand dollars or, where there is evidence of the street value of the dangerous drug, three times the street value of the dangerous drug, whichever is greater, and to imprisonment for a term which shall not be less than twenty-five years.
 - No persons were charged or convicted under sections 7, 8, or 9 in 2015 or 2016.
- 11. A medical practitioner who prescribes, administers, gives, sells or furnishes a dangerous drug to any person; or signs any prescription or order for a dangerous drug is guilty of an offence unless the dangerous drug is required for medicinal purposes or is prescribed for the medical treatment of a person who is under professional treatment by a medical practitioner.
 - No medical practitioner was convicted in contravention of the Dangerous Drugs Act for 2015 or 2016.
- 12. (1) The Minister may, by Order, appoint any person to be an inspector for the purposes of this Act.
 - No new inspectors have been appointed.
 - There are four inspectors attached to the Drug Inspectorate, namely:

Andrea Goddard-Grimes
 Mary Churaman-Lewis
 Jennifer Rodriguez-Herbert
 Pharmacist IV Ag. Principal Pharmacist IV
 Pharmacist III Ag. Pharmacist IV

4. Roxanne Rampersad Pharmacist III

Part IV: Summary Convictions

• No information is currently available.

Part V: Confiscation of Proceeds of Drug Trafficking

35D. (8) In this section "the prescribed sum" means such sum in Trinidad and Tobago currency as may be prescribed for the purposes of this section by an Order made by the Minister to whom responsibility for National Security is assigned and in determining under this section whether an amount of currency other than Trinidad and Tobago currency is not less than the prescribed sum that amount shall be converted at the prevailing rate of exchange.

• No information is currently available.

Part VIIA: Offences on the High Seas

53C. (2) The powers referred to in subsection (1) shall not be exercised outside the landward limits of the archipelagic waters of Trinidad and Tobago in relation to a ship registered in a Convention State except with the authority of the Minister to whom responsibility for Foreign Affairs is assigned, and he shall not give his authority unless that State has in relation to that ship—

- (a) requested the assistance of Trinidad and Tobago for the purpose mentioned in subsection (1); or (b) authorised Trinidad and Tobago to act for that purpose.
 - No requests were submitted to the Minister of Foreign and CARICOM Affairs for him to act under Section 53 C, Part VIII A of the Dangeroud Drugs Act, Chap 11:25 for 2015 and 2016.

Part IX: Miscellaneous

- 57. (2) The Minister of Finance may make Regulations for the seizure of any article or thing which is liable to forfeiture under this Act.
 - No new Regulations were made in 2015 or 2016.

Constraints and Limitations

The establishment of the Drug Inspectorate consists of eleven (11) pharmacists, eight (8) of whom are inspectors. In recent years however, the Drug Inspectorate has lost a significant number of pharmacists through attrition, voluntary retirement and resignation. This has resulted in a severely short-staffed department with only four (4) inspectors and no junior pharmacists. Succession planning has been virtually non-existent.

Furthermore, it is nearly impossible to attract and retain new pharmacists because of the compensation packages which are available in the public sector as compared to those offered in the private sector and some public health institutions. The present staffing constraints have resulted in the inability of the department to perform some of its critical functions efficiently and effectively.

This has negatively impacted several of the core activities of this department, including but not limited to the following:

- 1. The last decade has seen a rising trend in the incidence of unregistered and counterfeit narcotics being offered for sale at private pharmacies. This has been difficult to monitor and control due to the current staff constraints and consequent reduction in inspections.
- An increase in the availability of pharmacy services offered at public sector health facilities
 as well as the opening of new private sector pharmacies and medical clinics countrywide
 has led to an increased requirement for inspections which either could not be performed or
 completed in a timely manner.
- 3. General inspection audits of private and public pharmacies have declined considerably.
- 4. No inspectors were available to verify and destroy police narcotic exhibits for the years 2015 and 2016. This has been an ongoing problem for the last few years.

Additionally, it has been noted by the Inspectors that standards of practice, including record keeping, for pharmacies and pharmacists in the private sector are on the decline. This can in part be attributed to the infrequency of routine inspections by the Drug Inspectorate, which adversely affects the department's ability to adequately monitor the use of narcotics in the country.

Recommendations

The Drug Inspectorate Division has limited staff for the considerable amount of duties for which they are responsible. At present, the department is severely understaffed as seen in the table below:

TABLE OF COMPARISON OF ACTUAL STAFF VS ESTABLISHMENT POSTS

POST	ESTABLISHMENT	ACTUAL STAFF	REMARKS
Pharmacist I	3	0	
Pharmacist II	2	0	
Pharmacist III	6	1	3 Appointed, 2 Acting Pharmacist IV
Pharmacist IV	2	2	2 Acting as Pharmacist IV
Principal Pharmacist	1	1	

In order to mitigate the identified constraints, it is imperative that additional Pharmacists/Inspectors be appointed to the Drug Inspectorate so that the standards for the regulation of pharmacies and the monitoring and control of narcotics can be maintained and in some cases, improved.

Additionally, an update of the current legislation, including revision of pertinent regulations, is necessary for the efficient and effective control of dangerous drugs in the country.

Appendices

Appendix I: Applications Processed in 2015 and 2016.

Document	Quantity processed in 2015	Quantity processed in 2016
Licences to Sell and Distribute Narcotics	445	452
Licences to Import Narcotics	100	96
Certificate of Official Approval of Import for Narcotics	93	92
Authority for Withdrawal of Narcotics	80	72
No. of items imported	108	97

Appendix II: Reports Submitted to the International Narcotics Control Board (INCB) in 2015.

FORM	NAME OF FORM	Date submitted to INCB
Form A	Quarterly Statistics of Imports and Exports of Narcotic Drugs	4 th quarter 2014 26/02/2015 4 th quarter 2014 21/04/2015 (Amended) 1 st quarter 2015 21/04/2015; 2 nd quarter 2015 23/07/2015; 3 rd quarter 2015 26/10/2015
Form B	Annual Estimates of Requirements for Narcotic Drugs, Manufacture of Synthetic Drugs, and Cultivation of the Opium Poppy, the Cannabis Plant and the Coca Bush	For 2016 17/06/2015
Form C	Annual Statistics of Production, Manufacture, Consumption, Stocks and Seizures of Narcotic Drugs	For 2014 17/06/2015
Form D	Annual Information on Substances Frequently used in the Illicit Manufacture of Narcotic Drugs and Psychotropic Substances	For 2014 11/06/2015
Form P	Annual Statistical Report on Substances Listed in the Convention on Psychotropic Substances of 1971	No report submitted since no drugs listed in this Convention were imported or consumed

Appendix III: Reports Submitted to the International Narcotics Control Board (INCB) in 2016.

FORM	NAME OF FORM	Date submitted to INCB
Form A	Quarterly Statistics of Imports and Exports of Narcotic Drugs	4th quarter 2015 27/01/2016 1st quarter 2016 28/04/2016 2nd quarter 2016 22/07/2016 3rd quarter 2016 26/10/2016
Form B	Annual Estimates of Requirements for Narcotic Drugs, Manufacture of Synthetic Drugs, and Cultivation of the Opium Poppy, the Cannabis Plant and the Coca Bush	For 2017 09/06/2016
Form C	Annual Statistics of Production, Manufacture, Consumption, Stocks and Seizures of Narcotic Drugs	For 2015 09/06/2016
Form D	Annual Information on Substances Frequently used in the Illicit Manufacture of Narcotic Drugs and Psychotropic Substances	For 2015 29/06/2016
Form P	Annual Statistical Report on Substances Listed in the Convention on Psychotropic Substances of 1971	No report submitted since no drugs listed in this Convention were imported or consumed

Appendix IV: Licences to Import Narcotics Issued in 2015

NO.	LIC#	NAME OF FIRM	NO. OF	REMARKS
001	001/2015	A A Loguis Limited	ITEMS	CANCELLED
001	001/2015	A.A. Laquis Limited	2	CANCELLED
002	002/2013	Smith Robertson & Company Limited	2	
003	003/2015	Smith Robertson & Company	2	
003	003/2013	Limited	2	
004	004/2015	Alstons Marketing Company	2	
		Limited		
005	005/2015	A.A. Laquis Limited	1	
006	006/2015	A.A. Laquis Limited	1	
007	007/2015	Smith Robertson & Company	1	
		Limited		
008	008/2015	Smith Robertson & Company	2	
		Limited		
009	009/2015	Smith Robertson & Company	1	
010	010/2015	Limited	1	
010	010/2015	Smith Robertson & Company Limited	1	
011	011/2015	A.A. Laquis Limited	1	
011	011/2015	Bryden pi Limited	1	
013	012/2015	Bryden pi Limited Bryden pi Limited	1	
013	014/2015	Bryden pi Limited	1	CANCELLED
015	015/2015	Bryden pi Limited	2	CANCELLED
016	016/2015	Bryden pi Limited	1	OT IT (CELEED
017	017/2015	Smith Robertson & Company	2	
0 - 1		Limited	_	
018	018/2015	Physicians Pharmaceuticals	1	
		Limited		
019	019/2015	Smith Robertson & Company	1	
		Limited		
020	020/2015	Alstons Marketing Company	2	
0.5.1		Limited		
021	021/2015	A.A. Laquis Limited	1	
022	022/2015	C.L. Blanchfield & Company	1	
022	002/2015	Limited	1	
023	023/2015	A.A. Laquis Limited	1	
024	024/2015	A.A. Laquis Limited Trinidad and Tobago Forensic	1	
023	023/2013	Science Centre	1	
026	026/2015	Smith Robertson & Company	3	
020	020/2013	Limited]	
027	027/2015	Smith Robertson & Company	1	
	02772010	Limited		

NO.	LIC#	NAME OF FIRM	NO. OF ITEMS	REMARKS
028	028/2015	Trinidad and Tobago Forensic Science Centre	7	
029	029/2015	Alstons Marketing Company Limited	2	
030	030/2015	Bryden pi Limited	1	
031	031/2015	Bryden pi Limited	1	
032	032/2015	Smith Robertson & Company Limited	1	
033	033/2015	Alstons Marketing Company Limited	2	
034	034/2015	Smith Robertson & Company Limited	1	
035	035/2015	V & S Pharmaceuticals Limited	1	
036	036/2015	Bryden pi Limited	1	
037	037/2015	Bryden pi Limited	1	
038	038/2015	Bryden pi Limited	1	
039	039/2015	Bryden pi Limited	1	
040	040/2015	Bryden pi Limited	1	CANCELLED
041	041/2015	A.A. Laquis Limited	1	
042	042/2015	Smith Robertson & Company Limited	2	
043	043/2015	Smith Robertson & Company Limited	1	
044	044/2015	Smith Robertson & Company Limited	1	
045	045/2015	Smith Robertson & Company Limited	2	
046	046/2015	Smith Robertson & Company Limited	2	
047	047/2015	Smith Robertson & Company Limited	1	
048	048/2015	A.A. Laquis Limited	1	
049	049/2015	A.A. Laquis Limited	1	
050	050/2015	Alstons Marketing Company Limited	2	
051	051/2015	Smith Robertson & Company Limited	2	CANCELLED
052	052/2015	Smith Robertson & Company Limited	1	
053	053/2015	A.A. Laquis Limited	1	
054	054/2015	Bryden pi Limited	1	
055	055/2015	Bryden pi Limited	1	CANCELLED

NO.	LIC#	NAME OF FIRM	NO. OF ITEMS	REMARKS
056	056/2015	Alstons Marketing Company Limited	2	
057	057/2015	A.A. Laquis Limited	1	
058	058/2015	A.A. Laquis Limited	1	
059	059/2015	Smith Robertson & Company Limited	2	CANCELLED
060	060/2015	Smith Robertson & Company Limited	1	
061	061/2015	Smith Robertson & Company Limited	3	
062	062/2015	Smith Robertson & Company Limited	1	
063	063/2015	Bryden pi Limited	1	
064	064/2015	Trinidad and Tobago Forensic Science Centre	11	CANCELLED
065	065/2015	Smith Robertson & Company Limited	2	
066	066/2015	Smith Robertson & Company Limited	2	
067	067/2015	Smith Robertson & Company Limited	1	
068	068/2015	Bryden pi Limited	1	
069	069/2015	Alstons Marketing Company Limited	2	
070	070/2015	Alstons Marketing Company Limited	2	
071	071/2015	Physicians Pharmaceuticals Limited	1	
072	072/2015	Smith Robertson & Company Limited	1	
073	073/2015	Smith Robertson & Company Limited	2	
074	074/2015	Smith Robertson & Company Limited	2	
075	075/2015	Trinidad and Tobago Forensic Science Centre	11	
076	076/2015	Medi Supplies Limited	2	
077	077/2015	Medi Supplies Limited	2	CANCELLED
078	078/2015	Medi Supplies Limited	2	CANCELLED
079	079/2015	Medi Supplies Limited	2	CANCELLED
080	080/2015	Bryden pi Limited	1	
081	081/2015	Bryden pi Limited	1	

NO.	LIC#	NAME OF FIRM	NO. OF ITEMS	REMARKS
082	082/2015	C.L. Blanchfield & Company Limited	1	
083	083/2015	Smith Robertson & Company Limited	1	
084	084/2015	Smith Robertson & Company Limited	3	
085	085/2015	Alstons Marketing Company Limited	2	
086	086/2015	A.A. Laquis Limited	1	
087	087/2015	A.A. Laquis Limited	1	
088	088/2015	Alstons Marketing Company Limited	2	
089	089/2015	Smith Robertson & Company Limited	1	
090	090/2015	V & S Pharmaceuticals Limited	1	
091	091/2015	Smith Robertson & Company Limited	1	CANCELLED
092	092/2015	Smith Robertson & Company Limited	1	
093	093/2015	Smith Robertson & Company Limited	2	
094	094/2015	Alstons Marketing Company Limited	1	CANCELLED
095	095/2015	Smith Robertson & Company Limited	1	CANCELLED
096	096/2015	Smith Robertson & Company Limited	1	

Appendix V: Licences to Import Narcotics Issued in 2016

NO.	LIC#	NAME OF FIRM	NO. OF	REMARKS
110.		NAME OF FIRM	ITEMS	KEWAKI
001	001/2016	Smith Robertson & Company Limited	1	
002	002/2016	Alstons Marketing Company Limited	1	
003	003/2016	Smith Robertson & Company Limited	1	
004	004/2016	Smith Robertson & Company Limited	2	
005	005/2016	Alstons Marketing Company Limited	2	
006	006/2016	Smith Robertson & Company Limited	1	
007	007/2016	Smith Robertson & Company Limited	3	
008	008/2016	Smith Robertson & Company Limited	2	
009	009/2016	Ministry of Health	1	
010	010/2016	C.L. Blanchfield & Company Limited	1	
011	011/2016	Bryden pi Limited	1	
012	012/2016	Bryden pi Limited	1	
013	013/2016	Bryden pi Limited	1	
014	014/2016	Bryden pi Limited	1	
015	015/2016	Bryden pi Limited	1	
016	016/2016	Bryden pi Limited	1	
017	017/2016	Bryden pi Limited	1	
018	018/2016	Smith Robertson & Company Limited	1	
019	019/2016	Bryden pi Limited	1	
020	020/2016	Bryden pi Limited	1	
021	021/2016	Bryden pi Limited	1	CANCELLED
022	022/2016	Bryden pi Limited	1	
023	023/2016	Bryden pi Limited	1	
		Massy Distribution (Trinidad) a Division of		
024	024/2016	Massy Integrated Retail Limited	1	
		Massy Distribution (Trinidad) a Division of		
025	025/2016	Massy Integrated Retail Limited	1	
026	026/2016	Alstons Marketing Company Limited	2	
027	027/2016	A.A. Laquis (Trinidad) Limited	1	
028	028/2016	A.A. Laquis (Trinidad) Limited	1	
029	029/2016	A.A. Laquis (Trinidad) Limited	1	
030	030/2016	A.A. Laquis (Trinidad) Limited	1	
031	031/2016	Smith Robertson & Company Limited	1	
032	032/2016	Smith Robertson & Company Limited	1	
033	033/2016	Smith Robertson & Company Limited	1	
034	034/2016	Smith Robertson & Company Limited	1	
035	035/2016	Smith Robertson & Company Limited	1	
036	036/2016	Smith Robertson & Company Limited	1	
037	037/2016	Alstons Marketing Company Limited	2	
038	038/2016	Alstons Marketing Company Limited	2	
039	039/2016	Smith Robertson & Company Limited	1	
040	040/2016	Physicians Pharmaceuticals Limited	1	

NO.	LIC#	NAME OF FIRM	NO. OF ITEMS	REMARKS
041	041/2016	Bryden pi Limited	1	
042	042/2016	Bryden pi Limited	1	CANCELLED
043	043/2016	NIPDEC – Central Stores	1	CANCELLED
044	044/2016	NIPDEC – Central Stores	2	CANCELLED
045	045/2016	A.A. Laquis (Trinidad) Limited	1	
046	046/2016	Smith Robertson & Company Limited	1	
047	047/2016	Smith Robertson & Company Limited	1	
048	048/2016	A.A. Laquis (Trinidad) Limited	1	
049	049/2016	A.A. Laquis (Trinidad) Limited	1	
050	050/2016	Alstons Marketing Company Limited	2	
051	051/2016	NIPDEC – Central Stores	1	
052	052/2016	NIPDEC – Central Stores	2	
053	053/2016	Smith Robertson & Company Limited	2	
054	054/2016	A.A. Laquis (Trinidad) Limited	1	
055	055/2016	A.A. Laquis (Trinidad) Limited	1	
056	056/2016	Smith Robertson & Company Limited	1	
057	057/2016	Smith Robertson & Company Limited	1	
058	058/2016	Smith Robertson & Company Limited	3	
059	059/2016	Bryden pi Limited	1	CANCELLED
060	060/2016	Alstons Marketing Company Limited	2	
061	061/2016	Smith Robertson & Company Limited	1	
062	062/2016	Smith Robertson & Company Limited	3	
063	063/2016	Smith Robertson & Company Limited	1	
064	064/2016	Smith Robertson & Company Limited	2	
065	065/2016	Smith Robertson & Company Limited	3	
066	066/2016	Bryden pi Limited	1	
		Trinidad and Tobago Forensic Science		
067	067/2016	Centre	4	
068	068/2016	Alstons Marketing Company Limited	2	
069	069/2016	Alstons Marketing Company Limited	2	
070	070/2016	C.L. Blanchfield & Company Limited	1	
071	071/2016	Smith Robertson & Company Limited	1	
072	072/2016	Bryden pi Limited	1	
073	073/2016	Bryden pi Limited	1	
074	074/2016	Smith Robertson & Company Limited	1	
075	075/2016	Alstons Marketing Company Limited	2	
076	076/2016	A.A. Laquis (Trinidad) Limited	1	
077	077/2016	A.A. Laquis (Trinidad) Limited	2	
078	078/2016	Medi Supplies Limited	2	
079	079/2016	Medi Supplies Limited	2	
080	080/2016	Medi Supplies Limited	2	
081	081/2016	A.A. Laquis (Trinidad) Limited	1	
082	082/2016	Smith Robertson & Company Limited	1	

NO.	LIC#	NAME OF FIRM	NO. OF ITEMS	REMARKS
083	083/2016	Smith Robertson & Company Limited	2	
084	084/2016	Smith Robertson & Company Limited	1	
085	085/2016	Smith Robertson & Company Limited	1	
086	086/2016	A.A. Laquis (Trinidad) Limited	1	
087	087/2016	Smith Robertson & Company Limited	1	
088	088/2016	Alstons Marketing Company Limited	2	
089	089/2016	Smith Robertson & Company Limited	1	
090	090/2016	Smith Robertson & Company Limited	2	
091	091/2016	Smith Robertson & Company Limited	1	
092	092/2016	Smith Robertson & Company Limited	1	
093	093/2016	Smith Robertson & Company Limited	2	
094	094/2016	Alstons Marketing Company Limited	2	CANCELLED
095	095/2016	Smith Robertson & Company Limited	1	CANCELLED

Appendix VI: Licences to Sell and Distribute Narcotics Issued in 2015

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
001	001/2015	W.M. Allum's Pharmacy Limited	Kevin D. Lee
002	002/2015	Pyramid Drugs Limited	Trevor Moorley
003	003/2015	Numbers Drugs	Nazim Babwah
004	004/2015	A.A. Laquis Limited	Alicia Lara
005	005/2015	Hyper Pharm Limited	Jennifer Walters-Scott
006	006/2015	6 th Avenue Drugs	Neil Ahwai
007	007/2015	Eastside Pharmacy Limited	Nabeela Ramjohn
008	008/2015	Trinidad And Tobago Forensic Science	Arlette Lewis (Director)
	000,2010	Centre	1 11 1000 20 11 15 (2 11 00001)
009	009/2015	CVA Health and Wellness Croisee	Kellisha Antonia Hackett
010	010/2015	CVA Pharmacy	Courteney Augustine
011	011/2015	Seeta's Pharmacy	Dhanrajie Ragoonanan
012	012/2015	Rio Claro Pharmacy 2007 Limited	Veena Maharaj
013	013/2015	C & M Pharmacy Limited	Chandarbhan Maharaj
014	014/2015	Charis Drugs Limited	Sharmila Bachan
015	015/2015	Starlite Group Limited trading as Starlite	Carmen Abdelnor
		Pharmacy	
016	016/2015	Starlite Group Limited trading as Starlite	Jasmine Ali
		Drugs	
017	017/2015	U.W.I. Pharmacy	Andhra Maraj
018	018/2015	A.L.S. Drugs	Allison Patraj-Paray
019	019/2015	Drugs World Limited	Seunath Deonarine
020	020/2015	NSP Drug Mart	Nora Sookhoo-Persad
021	021/2015	Oscar Francois Limited	Neezamudin Khan
022	022/2015	Bryden pi Limited	Maria Jackson
023	023/2015	Clinicare Drugs Limited	Shiva Justin Singh
024	024/2015	Shiva's Pharmacy Limited	Shivanand Ramlogan
025	025/2015	Green Leaf Pharmacy Limited	Maida Lalic-Samaroo
026	026/2015	South Zone Pharmacy Limited	Ayoub Ali
027	027/2015	Farro's Pharmacy Limited	Robin Khan
028	028/2015	Young's Pharma-Serv Limited (Wholesale)	Alan Young
029	029/2015	Young's Pharma-Serv Limited	Alan Young
030	030/2015	Pharmax Pharmacy	Yasick Ali
031	031/2015	The Pharmacy	Nandaranie Singh
032	032/2015	KB's Pharmacy	Nigel Biersay
033	033/2015	WestShore Pharmacy	Kathy-Anne Gonzales
034	034/2015	Fit For Life Pharmacy Limited	Robert Soogrim
035	035/2015	Harrow's Drugs Company Limited	Herman Deonarine
036	036/2015	K. Pharm Limited	Ganesh Harrinanan
037	037/2015	Smith Robertson and Company Limited	Vishal Singh
038	038/2015	Smith Robertson and Company Limited	Carole Anne Maharaj
039	039/2015	Smith Robertson and Company Limited	Nisha Ramsundar
040	040/2015	Smith Robertson and Company Limited	Premela Manohar Mohammed

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
041	041/2015	Massy Distribution (Trinidad) a Division of	Tommy Luk Pat
		Massy Integrated Retail Limited	-
042	042/2015	Medical Associates Hospital Limited	Dr. Aroon Naraynsingh
		-	(Medical Doctor)
043	043/2015	Massy Stores Pharmacy	Meera Singh
044	044/2015	Massy Stores Pharmacy	Doodnath Mohammed
045	045/2015	Massy Stores Pharmacy	Charisse Christy Ramdeen
046	046/2015	Massy Stores Pharmacy	Majorie Philip
047	047/2015	Oscar Francois Limited	Kenny Arrindell
048	048/2015	Trinpharm Company Limited	Mitchell Ramsahai
049	049/2015	West Shore Medical Limited	Sree Venu Potluri (Managing
			Director)
050	050/2015	Medi Supplies Limited	Karen Teemal
051	051/2015	Centre Plus Pharmacy Limited	Nishanie Mahabir
052	052/2015	Benmah Drug Mart	Joann Jordan
053	053/2015	SuperPharm Lmited	Randy Ramoutar
054	054/2015	Ramsaran's Marketing and Services	Adita Maharaj
		Limited	
055	055/2015	Raawsk Limited trading as Southern Drug	Wendy Dookeran Diljohn
		Mart	
056	056/2015	Raawsk Limited trading as Southern Drug	Denzil Ramdass
		Mart	
057	057/2015	Oscar Francois Limited	James Mc Carthy
058	058/2015	Raman's Pharmacy Limited	Pradeep Raman
059	059/2015	Tiaren Drugs	Selvon Lum Kin
060	060/2015	Mickey's Pharmacy	Amjad I. Mohammed
061	061/2015	New City Drugs	Charles Cheeranjie
062	062/2015	Pharmacy World Limited	Devanand Ramdial
063	063/2015	Penny Med	Jimmy B. Cosico
064	064/2015	Pills and Beyond Pharmacy Limited	Cherisse Mootoosingh
065	065/2015	Mayaro Family Medicine	Joseph Ramsaroop
066	066/2015	The New Drug Store Limited	Mar-Reen Guevara
067	067/2015	Kappa Drugs	Andrew Sam
068	068/2015	De-Ann Drugs Limited	Soonya Ramraj
069	069/2015	Quick Check Pharmacy Limited	Nary Gunness
070	070/2015	CG's Pharmacy Limited	Chandra Gangoo
071	071/2015	Chaguanas Drug Mart Limited	Ken Baiju
072	072/2015	Felicity Community Pharmacy	Nigel Harnarine
073	073/2015	Infinity Drugs Limited	Safraz Mohammed
074	074/2015	Alstons Marketing Company Limited	Colin Inniss
075	075/2015	CVA Pharmacy (Wholesale)	Courteney Augustine
076	076/2015	Sav-Mor Drug Mart Limited	Vijay Soodeen
077	077/2015	Sinanan Drug World Limited	Arishma Y. Sinanan
078	078/2015	Shireen's Pharmacy	Yasmin S. Baksh

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
079	079/2015	Bryden pi Limited	Carin Flores
080	080/2015	Monarch Drugs	Ann Mohansingh
081	081/2015	Northgate Pharmacy Limited	Ngozi Efienemokwu
082	082/2015	Med-Zone Pharmacy Limited	Annika Anushka Sankar
083	083/2015	Pharm Sav Drugs Limited	Nadia Bhaggan
084	084/2015	Med-X Drugs Limited	Winstondave O'nardex
085	085/2015	Pleasantville Pharmacy Limited	Richard Owen John
086	086/2015	C.L. Blanchfield and Company Limited	Iruma Alicia Blanchfield
			(Chemical Engineer)
087	087/2015	Massy Distribution (Trinidad), A Division	Katyane Boland-Ramnarine
		Of Massy Integrated Retail Limited	
088	088/2015	St. Joseph Pharmacy	Shyam Ramoutar
089	089/2015	Tara's Pharmacy Limited	Anil Kumar Reddy Vaka
090	090/2015	R & R Budget Drugs Limited	Roann Persad
091	091/2015	Ron's Pharmacy Limited	Ria Doolarchan
092	092/2015	Southern Medical Services Company	Debera Lackhan-Maraj
		Limited	
093	093/2015	Tobago Pharmacy	Sandra Sinanan-Persad
094	094/2015	Triple K Pharmacy Limited	Kazim Ramjohn
095	095/2015	Tony's Drug Mart Limited	Anthony Sukhbir
096	096/2015	Vitadrugs Limited	Boodhan Ramgoolie
097	097/2015	Sharas Pharmacy Limited	Zorida Deopersad
098	098/2015	Wellmart Rx Limited	Joy Marie Eduardo
099	099/2015	Umraw's Pharmacy	Carol Umraw-Latchman
100	100/2015	Roopnarine's Drug Store Limited	Reann Roopnarine
101	101/2015	Mid-R Drugs	Rudranath Gangasingh
102	102/2015	Massy Stores Pharmacy	Miriam Gordon
103	103/2015	Prescription Plus Drugs Limited	Savera Ramsingh
104	104/2015	Medcorp Limited St. Clair Medical Centre	Joanne Calamba Cainto
105	105/2015	Community Joint Pharmacy	Iqbal Mohammed
106	106/2015	Valini's Drug Mart Limited	Valmiki Ramnarine
107	107/2015	Shalisa Pharmacy Limited	Elisabeth Thornhill
108	108/2015	Value Plus Pharmacy Limited	Cristal Sooknanan
109	109/2015	Health Wise Pharmacy Limited	Donna-Marie Bahadoor
110	110/2015	P.N.T. Drugs Limited	Taramatee Ruth Persad-
			Moonasar
111	111/2015	Diamond Vale Drugs Limited	Margarete Chang Kit
112	112/2015	Wazir's Pharmacy	Wazir Hosein
113	113/2015	Ralph's Pharmacy	Ralph Tilluckdharry
114	114/2015	Balmain Drug Mart	Curtis Victor Young Lum Kin
115	115/2015	Roopnarine's Pharmacy	Valerie Vedawatee
11.5	445/201=	Di Mari i i	Roopnarine
116	116/2015	Pharma KC Limited	Kimberly Abdul
117	117/2015	Sain's Pharmacy Limited	Vindra Ramsaran

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
118	118/2015	Ravi's Pharmacy	Ravi Deonarine
119	119/2015	Centra-Med Limited	Issa Ibn Abrahim Ali
120	120/2015	Carol's Pharmacy	Carissa Ganessingh
121	121/2015	C.T. Pharmacy Limited	Crystal Nalini Tahal
122	122/2015	Baggies Pharmacy Limited	Harold K. Bhagwandass
123	123/2015	GlaxoSmithKline Caribbean Limited	Jasmine Walker
124	124/2015	Bonanza (Investments) Drugs Limited	Rabidatt Bhaggan
125	125/2015	Easipharm Limited	Tricia Brathwaite
126	126/2015	Chase Budget Drugs Limited	Surash Harripersad
127	127/2015	A. Sinanan Drug Mart	Anthony Sinanan
128	128/2015	Capital Drugs Limited	Nolan Jordan
129	129/2015	Cumberdale Pharmacy	Kenrick Berkley Cumberbatch
130	130/2015	SuperPharm Limited	Su-Ann Joseph
131	131/2015	Yzee Drugs	Hamawatie Singh
132	132/2015	Health Site Pharmacy Company Limited	Vishnu Narine
133	133/2015	Health Mart Pharmacy	Sheryll Ann Balliramsingh
134	134/2015	Handels Pharmacy Limited	Handel Ali
135	135/2015	H & R Drugs	Kenny Jaipersad
136	136/2015	Gabriel's Pharmacy	Curtis Dann
137	137/2015	Real Friends Pharmacy Limited	Afroza Bhanji
138	138/2015	JGS Pharmacy Limited	Lester Albert
139	139/2015	JRC Discount Pharmacies	Margaret Isaac
140	140/2015	Jag's Drugs	Chandrika Jaggernath
141	141/2015	My Medicine Bag Limited	Natasha Nanan-Rampersad
142	142/2015	Midtown Pharmacy Limited	Sudarshan Devarashetty
143	143/2015	Medicine Corner Limited	Neil Varma Persad
144	144/2015	Nancy's Pharmacy Limited	Natasha Singh-Ramsook
145	145/2015	Tara's Pharmacy Limited	Michelle Ramlakhan
146	146/2015	First Health Pharmacy Limited	Latchman Samaroo
147	147/2015	Genethics Pharmaceuticals Limited	Valini Ramcharan
148	148/2015	Lakhan's Drugs	Kenneth Lakhan
149	149/2015	Community Hospital of Seventh Day Adventists	Jassodra Ramgoolam
150	150/2015	Barrackpore Medical Centre and Private	Zahirudeen Khan
		Hospital Limited	
151	151/2015	Health Plus Pharmacy Limited	Shobha Rampersadsingh
152	152/2015	Roystonia Pharmacy	Sumati Bakay
153	153/2015	Police Health Facility	Marcelle Charles
154	154/2015	Phamacare Drugs Limited	Jairam Maharaj
155	155/2015	Eastern Pharmacy	Keshwar Maharaj
156	156/2015	Cindy's Drugs Limited	Lilawattee Jane Bhagwandin
157	157/2015	Complete Care Pharmacy Services	Roger Cain
158	158/2015	BJ's Drug World	Ronald Joseph
159	159/2015	B & S Drugs	Candice Lalla

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
160	160/2015	D.M.C. Drugs	Kirk Mohan
161	161/2015	Express Drugs Limited	Jennifer Ragobar
162	162/2015	Physicians Pharmaceuticals Limited	Amisha Mohammed
163	163/2015	Anthony P. Scott and Company Limited	Amit Bhalekar
164	164/2015	Crichlow's Pharmacy	Carlhern Crichlow
165	165/2015	Bluestar Drugs	Marlene Ali-Makhan
166	166/2015	Croisee Pharmacy Limited	Rayavaram Siva Kumar
		·	Reddy
167	167/2015	D Village Pharm Limited	Rhea Ragobar
168	168/2015	Cunupia Drug Stop	Gobin Gopaul
169	169/2015	Jet Drugs	Aneil Narendra Ramnarace
170	170/2015	Healing Garden Pharmacy	Anita Ramkelawan
171	171/2015	Feel Good Drugs Limited	Ria Dhanoolal
172	172/2015	Family Care Drug Mart Limited	Hassanali Soobrattie
173	173/2015	Fabian's Pharmacy	Saeeda Ali
174	174/2015	Edoo's Drugs Limited	Mozam Edoo
175	175/2015	Ferrisa's Pharmacy	Fayad Mohammed
176	176/2015	Edinburgh Pharmacy Limited	Ravi Sobie
177	177/2015	Freeport Drugs Limited	Sharma Jaggernauth
178	178/2015	Garcia's Drugs Limited	Felix Stephen Garcia
179	179/2015	P.S.I. Drugs	Indra Pamela Mohammed
180	180/2015	Massy Stores Pharmacy	Navindra S. Persaud
181	181/2015	Ultra Care Pharmacy	Elni E. Samoramos
182	182/2015	Rite-Aid Pharmacy Limited	Dennis Ramkoon
183	183/2015	Ali's Pharmacy Limited	Michael Allum
184	184/2015	Ali's Pharmacy Limited	Shaun Vincent
185	185/2015	Eddie's Pharmacy Limited	Deonarine Mayrhoo
186	186/2015	Princes Town Pharmacy Limited	Karina Lutchmeesingh
187	187/2015	Horace's Drugs	Horace Ramjattan
188	188/2015	K.P.R. Pharmacy Limited	Joy-Roma Santo
189	189/2015	KAF Enterprises Limited trading as Harris	Nagasreenivas Musunuru
		Drugs	
190	190/2015	Jevestus Drugs	Claudette Browne
191	191/2015	Ali's Pharmacy Limited	Maurice Douglas
192	192/2015	Lisa's Pharmacy	Vandita Bissoonlal-Dubrey
193	193/2015	Ali's Pharmacy Limited	Prunella Watson-Woods
194	194/2015	Pharmtech Caribbean Limited	Caroline Maharaj
195	195/2015	Ali's Pharmacy Limited	Meela Ali
196	196/2015	Samodee's Drugstore	Farah Yassem
197	197/2015	Lisa's Pharmacy	Leah Sankarsingh-Ramnarine
198	198/2015	Lisa's Pharmacy Limited	Jacqueline Jeffers-Ramjohn
199	199/2015	Lisa's Drug Mart	Gale Ramsumair
200	200/2015	Live Well Pharmacy	Renuka Abdul
201	201/2015	Maraval Drugs Limited	Olusola Opeoluwa

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
202	202/2015	Medi-Buys Distribution Company Limited	Sherwin A. Narine
203	203/2015	Medsaver Plus Pharmacy Limited	Kimberly Kissoon
204	204/2015	Medicare Drugs (1990) Limited	Wendy Dawn Patel
205	205/2015	Med-Farm Drugs Limited	Juliet Marbella
206	206/2015	Med-Farm Drugs Limited	Shalima Mohammed
207	207/2015	Mc Bean Pharmacy	Neela Boodoo
208	208/2015	SuperPharm Limited	Elizabeth Akeung-Ho
209	209/2015	Miracle Pharmacy	Angela Dookeeram
210	210/2015	St. Augustine Private Hospital Pharmacy	Cindy K. Roopnarine-Dodoll
211	211/2015	Vernon Technologies Limited	Brenda Charles
212	212/2015	Reliant Pharmacy	Ronald Tsoi-A-Fatt
213	213/2015	Rite Health Pharmacy	Kavita Singh
214	214/2015	Ready- Pharm Limited	Afeefa Ali
215	215/2015	Gulf View Pharmacy Limited	Satis Maharaj
216	216/2015	Rajaram's Pharmacy	Ray Rajaram
217	217/2015	Couva Pharm Limited	Cindy Heeralal
218	218/2015	Couva Pharm Limited	Shazim Tullah
219	219/2015	Charlieville Community Pharmacy Limited	Shireen Ali
220	220/2015	Bhagan's Drugs Woodbrook Limited	Maria Linda Del Mundo
221	221/2015	Quality Care Pharmacy	Narindra Maharaj
222	222/2015	R & J Drugs Limited	Judy Naraynsingh
223	223/2015	Convenience Drug Mart Limited	Nal Pooransingh
224	224/2015	Pharmacy Plus	Kerra Dindial
225	225/2015	Pharmacy & Beyond Limited	Vedwattee Ramsaran
226	226/2015	Pharma-Choice Limited	Rishma Doodh
227	227/2015	MWL Incorporated Limited trading as The	Dinesh Moonan
		Drug Store, Rio Claro	
228	228/2015	New Age Pharmacy	Avinash Garib
229	229/2015	Montrose Pharmacy	Leanna David
230	230/2015	Mega Drugs Limited	Laila Seecharan
231	231/2015	Drugs R' Us Limited	Reah Toledo
232	232/2015	Karimed Pharmaceuticals Limited	Claire Crevier
233	233/2015	Massy Stores Pharmacy	Jaaved Mohammed
234	234/2015	SuperPharm Limited	Kavita Debysingh
235	235/2015	SuperPharm Limited	Nandini Ramsumair
236	236/2015	SuperPharm Limited	Danielle Gale
237	237/2015	Narinesingh's Drugs	Gopaul Narinesingh
238	238/2015	San Juan Pharmacy Limited	Khemraj Vishal Lakhan
239	239/2015	Sinanan's Drug Stop	Carlyle Sinanan
240	240/2015	Specialist Pharmacy Limited	Indira Singh-Bhagaloo
241	241/2015	Kadmin's One Stop Pharmacy Limited	Curtis Aimable
242	242/2015	SuperPharm Limited	Nadine Geoffroy
243	243/2015	V & C Drug Mart Limited	Kishore Ramcharan
244	244/2015	V & H Pharmacy	Gyandeo Shukla

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
245	245/2015	Unique Pharmacy	Varma Jules Boodhai
246	246/2015	Tunapuna Pharmacy and Cosmetic Centre	Brent Roopchandsingh
		Limited	
247	247/2015	Williamsville Drug Mart Limited	Kamie K. Ramkissoon
248	248/2015	Charlotte Drugs Limited	Addison Jagessar
249	249/2015	Vasans Pharmacy Limited	Indrani Rampersad-Mahabir
250	250/2015	Health Site Pharmacy Company Limited	Cherisse Ramsaran
251	251/2015	The Medical Dispensary	George Dottin
252	252/2015	Hometown Pharmacy Limited	Yvonne Greene
253	253/2015	Pill Dispenser Pharmacy Limited	Zahra Shah
254	254/2015	Curepe Pharmacy Limited	Nirvan Lall
255	255/2015	Tacarigua Pharmacy West	Terrence Gooding
256	256/2015	Heritage Pharmacy Limited	Aftard Abdool
257	257/2015	Solaris Publix Pharmacy Limited	Sherry-Ann Sinanan
258	258/2015	R & RS Drugs	Rahzia Sieuchand
259	259/2015	The Royal Pharmacy Limited	Allan Hewitt
260	260/2015	Tacarigua Pharmacy	Gokale Dial
261	261/2015	St. Helena Drug Mart	Nalini Dial
262	262/2015	A.D.C. Pharmacy	Rahaana Rajack
263	263/2015	P.C.L. Pharmacy	Glenwayne Suchit
264	264/2015	Rousillac Drug Mart	Tamara Benasrie
265	265/2015	Sinanan's Drug Store Limited	Sandra Sooknanan
266	266/2015	Tara's Pharmacy Limited	Suneel Kumar Thatiparthi
267	267/2015	Sun Crest International Limited	Brenda Grace Moore-
			Campbell
268	268/2015	SuperPharm Limited	Kimberley Ford
269	269/2015	Alpha Drugs Limited	Janet Emelda Gomes
270	270/2015	Southside Pharmacy	Llana Philip
271	271/2015	Med Smart Pharmacy Limited	Andrew Mohan
272	272/2015	Sav Mor Drug Mart Limited	Danushka Indarsingh
273	273/2015	Livingston Drugs Limited	Joslin Brewster
274	274/2015	Pharmacy Express Limited	Azhar Sajda Ali
275	275/2015	Glendale's Drugs	Glen Sewlal
276	276/2015	Yogi Pharm	Yogitaa Ragoonath
277	277/2015	Whole Life Pharmacy Limited	Alwyn Samuel Noble
278	278/2015	Bhagwat's Pharmacy	Leela Bhagwat Rajcoomar
279	279/2015	City Drugs Investments Limited	Wilson D. Reyes
280	280/2015	Chaguanas Pharmacy Limited	Shantal Boodoo
281	281/2015	Ramkoon's Pharmacy	Desmond Ramkoon
282	282/2015	The Surgi-Med Clinic	Dr. Anirudh Mahabir
			(Ophthalmologist)
283	283/2015	Saddle Pharmacy	Carmel Travero
284	284/2015	Personal Care Pharmacy Limited	Marissa Naidoo
285	285/2015	Palmyra Pharmacy Limited	Ranjit Seegobin

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
286	286/2015	Lyn's Pharmacy Limited	Carolyn Agnoo Maharaj
287	287/2015	Lyn's Pharmacy Limited	Khadija Dorset
288	288/2015	Hilltop Pharmacy Limited	Ricardo Mohammed
289	289/2015	Palmyra Pharmacy Limited	Morisha Manick
290	290/2015	The Family Medicine Centre Limited	Aneil Hosein
291	291/2015	The Friendly Pharmacy Limited trading as	Lutchman Lalchan
		St. James Pharmacy	
292	292/2015	Waterfront Pharmacy	Auldith Beache
293	293/2015	Verison Drugs Limited	Patrick Harris
294	294/2015	Universal Pharmacy	Marlon Austin
295	295/2015	You and Us Pharmacy Limited	Michael Awai
296	296/2015	V & S Pharmaceuticals Limited	Varune Persad
297	297/2015	DR Pharmacy Limited	Vijay Soodeen
298	298/2015	Roger's Drug Store Limited	Jefferson Val Bishop
299	299/2015	Farmacia Cruz Limited	Jose Dennis Cruz
300	300/2015	La Romain Drug Mart Limited	Kaloutie Singh
301	301/2015	Pharma Rx Pharmacy	Mathias Marfan
302	302/2015	Quik-Pharm	Cuthbert Ferdinand
303	303/2015	Lee's Pharmacy Limited	Miguel Bujhawan
304	304/2015	Lopinot Drugs Limited	Dineash Bidaisee
305	305/2015	Prescription House Limited	Sunita Hardit-Singh
306	306/2015	Kappa Drugs	Andrew Ramirez
307	307/2015	Randall's Pharmacy Limited	David Rahaman
308	308/2015	Harry's Pharmacy	Harry Kanta Persad
309	309/2015	Longdenville Pharmacy	Cheryl Vanda Gopaul
310	310/2015	Bhagan's Drugs (Price Plaza) Limited	Shane Rampersad
311	311/2015	Deniceia Pharmacy	Renuka Mahadeo
312	312/2015	Maxsal Drugs	Earl Fitzgerald Mason
313	313/2015	Bhagan's Drugs (Price Plaza) Limited	Shelly-Ann Ramsamooj
314	314/2015	City Centre Pharmacy Limited	Gail Ramnath
315	315/2015	Central Drugs Limited	Nizam Asmath
316	316/2015	Pharmacy Connection Limited	Navin Deonarine
317	317/2015	Park View Drugs	Aliyyah Mohammed
318	318/2015	Delta Drugs	Dale Sewlal
319	319/2015	Priority Drugs	Vashish Maharaj
320	320/2015	10 th Avenue Drugs Limited	Sonia Mc Quilkin
321	321/2015	Hetero Drugs	Murali Mohan Reddy
			Gogireddy
322	322/2015	Bournes Road Pharmacy	Andrew Rahaman
323	323/2015	Medisource Pharmacy Limited	Nadia Ramsawack
324	324/2015	Bhagan's Drugs (Price Plaza) Limited	Delita Dana Ragoonanan
325	325/2015	Carol's Pharmacy	Roseann St. Rose
326	326/2015	Shoprite Drugs	Atri Seepersad
327	327/2015	Port-Of-Spain General Hospital	Elson Robinson

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
328	328/2015	Disdrugs	Carissa Ganessingh
329	329/2015	Lisa's Pharmacy	Tricia Lerisa Tahal
330	330/2015	JRC Discount Pharmacies	Michelle Mohammed
331	331/2015	Pharmacy Plus	Anusha Beharrysingh
332	332/2015	Western Pharmacy (Walker's) Limited	Mary Jane S. Lamigo
333	333/2015	Jad's Pharmacy	Denise Mohammed
334	334/2015	Bhagan's Drugs (San Juan) Limited	Salina Dhanoolal
335	335/2015	Samodee's Drugstore	Shalini Mohanlal
336	336/2015	O'meara Pharmacy Limited	Milagros De Leon Acosta
337	337/2015	KW Pharmacy	Reanne Dinnoo
338	338/2015	City Centre Pharmacy Limited	Teresina Lucio
339	339/2015	Maxsal Drugs Limited	Shabiyath Ali
340	340/2015	M & R Pharmacy	Meera Heeraman
341	341/2015	Eye See You Ophthalmic and Medical	Cassie Joseph
		Supplies Limited	_
342	342/2015	Vark Enterprises Limited trading as	Anil Bheem
		Medicine Plus	
343	343/2015	Living Water Hospice	Richard Anthony Clerk
			(Medical Doctor)
344	344/2015	Bhagan's Drugs (Tobago) Limited	Amelia Birusingh-Cipriani
345	345/2015	Total Care Pharmacy Limited	Ava Ramsaroop
346	346/2015	The Medicine Stop	Judy Dass-Ali
347	347/2015	Easipharm Limited	Pamela Townsend
348	348/2015	Apollo Pharmacy Limited	Raja Sekhara Reddy
			Sanikommu
349	349/2015	Uptown Drugs Limited	Kamal Singh
350	350/2015	Wellness Pharmacy Limited	Steve Ganessingh
351	351/2015	Quik-Pharm	Alana Bosland
352	352/2015	We Care Pharmacy Limited	Rai Ragoonanan
353	353/2015	Bhagan's Drugs (Tobago) Limited	Max Ferdinand
354	354/2015	Busy Corner Drugs	Steve Ramlogan
355	355/2015	Williams Drug Store Limited	Trevor Williams
356	356/2015	Bhagan's Drugs (1996) Limited	Marvin Chotai
357	357/2015	Bhagan's Drugs (1996) Limited	Sean Bhagan
358	358/2015	Drugs & Beyond Limited	Savitri Sandra Ramnanan
359	359/2015	Eddie's Pharmacy Limited	Emerson Ramjattan
360	360/2015	Tropical Pharmacy Limited	Ricky Roodal
361	361/2015	AVS Pharmacy Limited	Vijay Pundit
362	362/2015	Feel Good Drugs Limited	Ria Dhanoolal
363	363/2015	Customer Care Pharmacy	Bibi Rafeena Goolab-
			Rampersad
364	364/2015	Victoria Nursing Home Limited	Dr. Clem Calvin Ragobar
			(Medical Practitioner)
365	365/2015	Island Drugs Pharmacy Limited	Bennett Oliver Kangalee

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
366	366/2015	Medac Drugs Limited	Bashaar Ahmed Khan
367	367/2015	Loren's Pharmacy Limited	Sherolyn Maharaj-Singh
368	368/2015	Princes Town Drug Mart Limited	Susan Khan
369	369/2015	A & N Drug Mart Limited	Anilla Ananda Ballyram
370	370/2015	A & L Pharmacy (2005) Limited	Clyde Albert Allum
371	371/2015	H & R Drugs	Subhash Bahadur Singh
372	372/2015	A & R Budget Drugs	Cecil Mierez
373	373/2015	Pharm Sav Drugs Limited	Ravindra Sookram
374	374/2015	Samz Pharmafit Limited	Samantha Rajkumar
375	375/2015	Petrotrin Medical Centre Dispensary	Ilva Crooks
		Pointe-A-Pierre	
376	376/2015	Clinicare Drugs Limited	Nadia Melissa Asgarali
377	377/2015	BJ's Drug World Limited	Ronald Joseph
378	378/2015	MWL Incorporated trading as The Drug	Arif Yasin Hosein-Shah
		Store, Marabella	
379	379/2015	Petrotrin Medical Centre Dispensary	Bertina Hernandez
		Guayaguayare	
380	380/2015	Petrotrin Medical Centre Dispensary Point	Hazel Keppel
		Fortin	
381	381/2015	Petrotrin Medical Centre Dispensary Penal	Christine Corbie
382	382/2015	Petrotrin Trinmar Operations Medical	Elizabeth Huggins-Andrews
		Centre	
383	383/2015	The Doctors' Clinic Net International	Dr. Robert Naidike (Surgeon
		Limited	And Physician)
384	384/2015	Advantus Management Limited trading as	Princess Denoon-King
20.5	205/2015	10 th Avenue, The Pharmacy	
385	385/2015	Sinanan's West End Pharmacy Limited	Christopher Sinanan
386	386/2015	Drugs 4 U Pharmacy	Phaedra Sookdeo
387	387/2015	Khan's Pharmacy	Ameena Ali
388	388/2015	Petrotrin Medical Centre Dispensary Santa	Diane Julien
200	200/2015	Flora	D. C. C. El.
389	389/2015	Petrotrin Medical Centre Dispensary	Patricia Elvin
200	390/2015	Brighton Petrotrin Medical Stores	Solver Com
390	390/2013		Selwyn Carr
391 392	391/2013	Rampersad Medical Centre Limited Saddle Pharmacy	Jason Rampersad Jerard Castillo
392	392/2013	Belmont Pharmacy	Edith Marcano-Smith
393	393/2013	· · · · · · · · · · · · · · · · · · ·	Daphne Bartlett
394	394/2013	D. Bartlett Drugs Limited Pharmsav Drugs Limited	Nadia Bhaggan
395	395/2015	Good Life Pharmacy	Carmel Travero
390	390/2013	Health Image Pharmacy	Sarvy Ramlogan
398	398/2015		Kala Budhooram
		CVA Pharmacy St. James	Kala Budhooram Kellisha Antonia Hackett
399	399/2015	CVA Pharmacy, St. James	
400	400/2015	Lays Drugs Limited	Alyssa Ramadar

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
401	401/2015	Med Smart Pharmacy Limited	Kenny Jaipersad
402	402/2015	Med-Farm Drugs Limited	Nellie Singh-Musai
403	403/2015	MK Pharmacy Limited	Milagros De Leon Acosta
404	404/2015	Jevestus Drugs	Claudette Browne
405	405/2015	Massy Distribution (Trinidad) a Division of	Shelley Cave Daisley
		Massy Integrated Retail Limited	
406	406/2015	Tara's Pharmacy Limited	Reanne Dinnoo
407	407/2015	Capital Drugs Limited	Pradeep Mallireddy
408	408/2015	Tablet City Limited	Carol Elizabeth Ramsoomair
409	409/2015	Macoya Pharmacy	Rawle Mooteeram
410	410/2015	Alstons Marketing Company Limited	Colin Inniss
411	411/2015	The Avenue Pharmacy	Tommy Luk Pat
412	412/2015	Bhagan's Drugs (Tobago) Limited	Vere Bhagan
413	413/2015	Carol's Pharmacy	Kinan Safi
414	414/2015	Eye See You Ophthalmic and Medical	Soraya Boodoo
		Supplies Limited	
415	415/2015	Water Sprout Pharmacy & Wellness Center	Denise Mohammed
416	416/2015	Raawsk Limited trading as Southern Drug	Farana Mohammed
		Mart	
417	417/2015	SuperPharm Limited	Denzil Ramdass
418	418/2015	Pennywise Pharmacy	Nal Pooransingh
419	419/2015	Drugs 4 U Pharmacy	Soonya Ramraj
420	420/2015	Couva Pharm Limited	Cindy Heeralal
421	421/2015	Massy Distribution (Trinidad) a Division of	Elizabeth Akeung-Ho
		Massy Integrated Retail Limited	
422	422/2015	Lisa's Pharmacy	Leah Sankarsingh-Ramnarine
423	423/2015	Family Care Pharmacy	Ernest Sankar
424	424/2015	Pharm-Aid Company Limited trading as	Rhonda Karim
		Apex Pharmacy	
425	425/2015	Kappa Drugs	Nicki De Bourg
426	426/2015	Massy Distribution (Trinidad) a Division of	Nisha Khan-Roopnarine
		Massy Integrated Retail Limited	
427	427/2015	SuperPharm Limited	Sharon Thomson
428	428/2015	Tara's Pharmacy Limited	Reanne Dinnoo
429	429/2015	Tara's Pharmacy Limited	Phaedra Sookdeo
430	430/2015	Pennywise Pharmacy	Rishi Persad
431	431/2015	Health Site Pharmacy Company Limited	Jerard Castillo
432	432/2015	Nancy's Pharmacy Limited	Natasha Singh Ramsook
433	433/2015	K.R.T. Drugs Plus	Lakshmi Lydia Maharaj
434	434/2015	Westmed Pharmacy	Melody Bay-An
435	435/2015	Pennywise Pharmacy	Nadia Ramsawack
436	436/2015	Petrotrin Augustus Long Hospital	Diane Julien
		Dispensary	

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
437	437/2015	Petrotrin Medical Centre Dispensary Forest	Kala Seereeram
		Reserve	
438	438/2015	Petrotrin Medical Centre Dispensary Santa	Jacqueline Ifill
		Flora	
439	439/2015	Massy Stores Pharmacy	Safiyya Ali-Mohammed
440	440/2015	Acropolis Medical Centre Limited	Jose Dennis Cruz
441	441/2015	SuperPharm Limited	Cherisse Ramsaran
442	442/2015	Anva's Health Plus Pharmacy	Vandita Bissoonlal-Dubrey
		Cancelled	
443	443/2015	Danny's Medicine Corner Limited	Shivanand Phagoo
		Cancelled	
444	444/2015	Health Central Limited Pharmacy	Kavita Dukharan
		Cancelled	
445	445/2015	Medi-Time Pharmacy Limited	Donna Naila Parey

Appendix VII: Licences to Sell and Distribute Narcotics Issued in 2016

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
001	001/2016	Farro's Pharmacy Limited	Robin Khan
002	002/2016	Pyramid Drugs Limited	Trevor Moorley
003	003/2016	A.A. Laquis (Trinidad) Limited	Alicia Lara
004	004/2016	Massy Stores Pharmacy	Navindra S. Persaud
005	005/2016	Massy Stores Pharmacy	Salma Khan
006	006/2016	Massy Stores Pharmacy	Doodnath Mohammed
007	007/2016	Bryden pi Limited	Carin Flores
008	008/2016	Bryden pi Limited	Maria Jackson
009	009/2016	Massy Stores Pharmacy	Jaaved Mohammed
010	010/2016	Massy Stores Pharmacy	Charisse Christy Ramdeen
011	011/2016	Starlite Group Limited trading as Starlite Drugs	Jasmine Ali
012	012/2016	Starlite Group Limited trading as Starlite Pharmacy	Carmen Abdelnor
013	013/2016	Oscar Francois Limited	Kenny Arrindell
014	014/2016	Oscar François Limited	Neezamudin Khan
015	015/2016	Pharmacy Express Limited	Azhar Sajda Ali
016	016/2016	Danny's Medicine Corner Limited	Shivanand Phagoo
017	017/2016	CG's Pharmacy Limited	Chandra Gangoo
018	018/2016	West Shore Medical Limited	Sree Venu Potluri (Managing Director)
019	019/2016	Centra-Med Limited	Issa Ibn Abrahim Ali
020	020/2016	The Pharmacy	Nandaranie Singh
021	021/2016	NSP Drug Mart	Nora Sookhoo-Persad
022	022/2016	Roopnarine's Pharmacy	Valerie Vedawatee
			Roopnarine
023	023/2016	Massy Stores Pharmacy	Majorie Philip
024	024/2016	WestShore Pharmacy	Kathy-Anne Gonzales
025	025/2016	<u> </u>	Carol Umraw-Latchman
026	026/2016	,	Sudarshan Devarashetty
027	027/2016	Medcorp Limited St. Clair Medical Centre	Joanne Calamba Cainto
028	028/2016	Cumberdale Pharmacy	Kenrick Berkley Cumberbatch
029	029/2016	Community Hospital of Seventh Day Adventists	Jassodra Ramgoolam
030	030/2016	C.L. Blanchfield and Company Limited	Iruma Alicia Blanchfield (Chemical Engineer)
031	031/2016	Rio Claro Pharmacy 2007 Limited	Veena Maharaj
032	032/2016	Handel's Pharmacy Limited	Handel Ali
033	033/2016	Harrow's Drugs Company Limited	Herman Deonarine
034	034/2016	Community Joint Pharmacy	Iqbal Mohammed
035	035/2016	Drugs 4 U Pharmacy	Soonya Ramraj
036	036/2016	B & S Drugs	Candice Lalla
037	037/2016	Infinity Drugs Limited	Safraz Mohammed

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
038	038/2016	Hyper Pharm Limited	Jennifer Walters-Scott
039	039/2016	Drugs R' Us Limited	Reah Toledo
040	040/2016	Alpha Drugs Limited	Janet Emelda Gomes
041	041/2016	Numbers Drugs	Nazim Babwah
042	042/2016	Smith Robertson and Company Limited	Premela Manohar Mohammed
043	043/2016	Smith Robertson and Company Limited	Vishal Singh
044	044/2016	Smith Robertson and Company Limited	Carole Anne Maharaj
045	045/2016	Smith Robertson and Company Limited	Nisha Ramsundar
046	046/2016	Massy Distribution (Trinidad) a Division of	Elizabeth Akeung-Ho
		Massy Integrated Retail Limited	
047	047/2016	Raman's Pharmacy Limited	Pradeep Raman
048	048/2016	JGS Pharmacy Limited	Lester Albert
049	049/2016	JRC Discount Pharmacies	Michelle Mohammed
050	050/2016	KB's Pharmacy	Nigel Biersay
051	051/2016	Centre Plus Pharmacy Limited	Nishanie Mahabir
052	052/2016	U.W.I. Pharmacy	Andhra Maraj
053	053/2016	Loren's Pharmacy Limited	Sherolyn Maharaj-Singh
054	054/2016	K. Pharm Limited	Ganesh Harrinanan
055	055/2016	Clinicare Drugs Limited	Shiva Justin Singh
056	056/2016	Eastside Pharmacy Limited	Nabeela Ramjohn
057	057/2016	Maxsal Drugs Limited	Shabiyath Ali
058	058/2016	Massy Distribution (Trinidad) a Division of	Katyane Boland-Ramnarine
		Massy Integrated Retail Limited	
059	059/2016	Trinpharm Company Limited	Mitchell Ramsahai
060	060/2016	Health Central Limited Pharmacy	Kavita Dukharan-Ramkay
061	061/2016	Medsaver Plus Pharmacy Limited	Kimberly Kissoon
062	062/2016	Mickey's Pharmacy	Amjad I. Mohammed
063	063/2016	Mid-R Drugs	Rudranath Gangasingh
064	064/2016	Miracle Pharmacy	Angela Dookeeram
065	065/2016	Quik-Pharm	Michael Abdullah
066	066/2016	Alstons Marketing Company Limited	Colin Inniss
067	067/2016	Lisa's Pharmacy	Amanda Caroline Espinoza
068	068/2016	Park View Drugs	Aliyyah Mohammed
069	069/2016	Monarch Drugs	Ann Mohansingh
070	070/2016	Northgate Pharmacy Limited	Ngozi Efienemokwu
071	071/2016	Lakhan's Drugs	Kenneth Lakhan
072	072/2016	Narinesingh's Drugs	Gopaul Narinesingh
073	073/2016	GlaxoSmithKline Caribbean Limited	Jasmine Walker
074	074/2016	CVA Pharmacy	Courteney Augustine
075	075/2016	CVA Pharmacy, St. James	Kellisha Antonia Hackett
076	076/2016	Deniceia Pharmacy	Ana-Rhea Mohamed-Hoosanie
077	077/2016	DR Pharmacy Limited	Deonarine Ramlal
078	078/2016	Tobago Pharmacy	Sandra Sinanan-Persad
079	079/2016	Wellmart Rx Limited	Joy Marie Eduardo

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
080	080/2016	Vitadrugs Limited	Boodhan Ramgoolie
081	081/2016	Acropolis Medical Centre Limited	Jose Dennis Cruz
082	082/2016	Benmah Drug Mart	Joann Jordan
083	083/2016	Carol's Pharmacy	Kinan Safi
084	084/2016	Barrackpore Medical Centre and Private	Zahirudeen Khan
		Hospital Limited	
085	085/2016	Mayaro Family Medicine	Joseph Ramsaroop
086	086/2016	CVA Pharmacy (Wholesale)	Courteney Augustine
087	087/2016	CVA Pharmacy	Elni E. Samoramos
088	088/2016	Clinicare Drugs Limited	Nadia Melissa Asgarali
089	089/2016	Triple K Pharmacy Limited	Kazim Ramjohn
090	090/2016	Charlieville Community Pharmacy Limited	Shireen Ali
091	091/2016	Central Drugs Limited	Nizam Asmath
092	092/2016	The New Drug Store Limited	Mar-Reen Josette Guevara
093	093/2016	Chaguanas Pharmacy Limited	Shantal Boodoo
094	094/2016	Oscar Francois Limited	James Mc Carthy
095	095/2016	Physicians Pharmaceuticals Limited	Amisha Mohammed
096	096/2016	Eye See You Ophthalmic and Medical	Soraya Boodoo
		Supplies Limited	
097	097/2016	Massy Stores Pharmacy	Safiyya Ali-Mohammed
098	098/2016	Sain's Pharmacy Limited	Vindra Ramsaran
099	099/2016	Ralph's Pharmacy	Ralph Tilluckdharry
100	100/2016	Rousillac Drug Mart	Tamara Benasrie
101	101/2016	Priority Drugs	Vashish Maharaj
102	102/2016	Pill Dispenser Pharmacy Limited	Zahra Shah
103	103/2016	Sharas Pharmacy Limited	Zorida Deopersad
104	104/2016	Shireen's Pharmacy	Yasmin S. Baksh-Shah
105	105/2016	Shalisa Pharmacy Limited	Elisabeth Thornhill-Rajpaul
106	106/2016	Sav-Mor Drug Mart Limited	Danushka Indarsingh
107	107/2016	Sinanan Drug World Limited	Arishma Y. Sinanan
108	108/2016	Truecare Pharmacy	Rishma Doodh
109	109/2016	O'meara Pharmacy Limited	Latchman Samaroo
110	110/2016	Heritage Pharmacy Limited	Aftard Abdool
111	111/2016	Med Smart Pharmacy Limited	Kenny Jaipersad
112	112/2016	Drugs World Limited	Seunath Deonarine
113	113/2016	Hilltop Pharmacy Limited	Ricardo Mohammed
114	114/2016	Western Pharmacy (Walker's) Limited	Mary Jane S. Lamigo
115	115/2016	Sun Crest International Limited	Brenda Grace Moore-
			Campbell
116	116/2016	Island Drugs Pharmacy Limited	Fazal Hosein
117	117/2016	BJ's Drug World Limited	Ronald Joseph
118	118/2016	Tiaren Drugs	Selvon Lum Kin
119	119/2016	Solaris Publix Pharmacy Limited	Sherry-Ann Sinanan
120	120/2016	Quick Check Pharmacy Limited	Nary Gunness

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
121	121/2016	Pharmax Pharmacy	Yasick Ali
122	122/2016	Tara's Pharmacy Limited	Anil Kumar Reddy Vaka
123	123/2016	Tara's Pharmacy Limited	Reanne Dinnoo
124	124/2016	Tara's Pharmacy Limited	Phaedra Sookdeo
125	125/2016	Wazir's Pharmacy	Wazir Hosein
126	126/2016	Anthony P. Scott and Company Limited	Amit Bhalekar
127	127/2016	Sinanan's West End Pharmacy Limited	Christopher Sinanan
128	128/2016	The Avenue Pharmacy	Tommy Luk Pat
129	129/2016	Avocat Pharm-Assist Drugs Limited	Margaret Isaac
130	130/2016	Cornerstone Pharmacy Limited	Kala Budhooram
131	131/2016	Raawsk Limited trading as Southern Drug Mart	Ahamad Mohammed
132	132/2016	Disdrugs	Carissa Ganessingh
133	133/2016	Medi-Time Pharmacy Limited	Donna N. Parey
134	134/2016	Curepe Pharmacy Limited	Nirvan Lall
135	135/2016	Pharma KC Limited	Kimberly Abdul
136	136/2016	New City Drugs	Charles Cheeranjie
137	137/2016	Pennywise Pharmacy	Rishi Persad
138	138/2016	Pennywise Pharmacy	Nadia Ramsawack
139	139/2016	W.M. Allum's Pharmacy Limited	Kevin D. Lee
140	140/2016	SuperPharm Limited	Kavita Debysingh
141	141/2016	SuperPharm Limited	Denzil Ramdass
142	142/2016	SuperPharm Limited	Samantha Keshwar
143	143/2016	Bhagan's Drugs Woodbrook Limited	Maria Linda Del Mundo
144	144/2016	Pennywise Pharmacy	Nal Pooransingh
145	145/2016	Shoprite Drugs	Atri Seepersad
146	146/2016	Pharmacy World Limited	Devanand Ramdial
147	147/2016	SuperPharm Limited	Kimberley Ford
148	148/2016	A. Sinanan Drug Mart	Anthony Sinanan
149	149/2016	A.L.S. Drugs	Allison Patraj-Paray
150	150/2016	AVS Pharmacy Limited	Vijay Pundit
151	151/2016	Pleasantville Pharmacy Limited	Richard Owen John
152	152/2016	Bhagan's Drugs (Price Plaza) Limited	Shelly-Ann Ramsamooj
153	153/2016	Bhagan's Drugs (Price Plaza) Limited	Delita Dana Ragoonanan
154	154/2016	Charis Drugs Limited	Sharmila Bachan
155	155/2016	Bhagan's Drugs (Price Plaza) Limited	Shane Rampersad
156	156/2016	Bhagwat's Pharmacy	Leela Bhagwat Rajcoomar
157	157/2016	Penny Savers Pharmacy Limited	Kreshma Roopnarine
158	158/2016	Montrose Pharmacy	Leanna David
159	159/2016	Medical Associates Hospital Limited	Dr. Aroon Naraynsingh
			(Medical Doctor)
160	160/2016	Advantus Management Limited trading as 10 th Avenue, The Pharmacy	Princess Denoon-King
161	161/2016	Jag's Drugs	Chandrika Jaggernath

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
162	162/2016	Kadmin's One Stop Pharmacy Limited	Curtis Aimable
163	163/2016	KAF Enterprises Limited trading as Harris	Nagasreenivas Musunuru
		Drugs	
164	164/2016	KPR Pharmacy Limited	Joy-Roma Santo
165	165/2016	Jet Drugs	Aneil Narendra Ramnarace
166	166/2016	Ministry of Health	Adelle-Lisa Chang On
			(County Medical Officer of
			Health)
167	167/2016	Multicare Pharmacy Limited	Alisha Aggan
168	168/2016	K.R.T. Drugs Plus	Lakshmi Lydia Maharaj
169	169/2016	SuperPharm Limited	Nadine Geoffroy
170	170/2016	SuperPharm Limited	Su-Ann Joseph
171	171/2016	Rajaram's Pharmacy	Ray Rajaram
172	172/2016	SuperPharm Lmited	Randy Ramoutar
173	173/2016	Yzee Drugs	Hamawatie Singh
174	174/2016	Randall's Pharmacy Limited	David Rahaman
175	175/2016	Ravi's Pharmacy	Ravi Deonarine
176	176/2016	Shiva's Pharmacy Limited	Shivanand Ramlogan
177	177/2016	Ron's Pharmacy Limited	Ria Doolarchan
178	178/2016	Ready- Pharm Limited	Afeefa Ali
179	179/2016	St. Augustine Private Hospital Pharmacy	Cindy K. Roopnarine-Dodoll
180	180/2016	Personal Care Pharmacy Limited	Marissa Naidoo
181	181/2016	Medicare Drugs (1990) Limited	Wendy Dawn Patel
182	182/2016	Medi-Buys Distribution Company Limited	Sherwin A. Narine
183	183/2016	Rite Health Pharmacy	Kavita Singh
184	184/2016	Raawsk Limited trading as Southern Drug	Wendy Dookeran Diljohn
		Mart	
185	185/2016	Real Friends Pharmacy Limited	Afroza Bhanji
186	186/2016	Princes Town Pharmacy Limited	Karina Lutchmeesingh
187	187/2016	P.N.T. Drugs Limited	Taramatee Ruth Persad-
			Moonasar
188	188/2016	Longdenville Pharmacy	Cheryl Vanda Gopaul
189	189/2016	Harry's Pharmacy	Harry Kanta Persad
190	190/2016	D.M.C. Drugs	Kirk Mohan
191	191/2016	Bonanza (Investments) Drugs Limited	Rabidatt Bhaggan
192	192/2016	Bhagan's Drugs (1996) Limited	Sean Bhagan
193	193/2016	R & R Budget Drugs Limited	Roann Persad
194	194/2016	Roopnarine's Drug Store Limited	Reann Roopnarine
195	195/2016	Health Wise Pharmacy Limited	Donna-Marie Bahadoor
196	196/2016	Massy Stores Pharmacy	Ravindra Sookram
197	197/2016	MK Pharmacy Limited	Milagros De Leon Acosta
198	198/2016	10 th Avenue Drugs Limited	Sonia Mc Quilkin
199	199/2016	Tony's Drug Mart Limited	Anthony Sukhbir
200	200/2016	The Family Medicine Centre Limited	Aneil Hosein

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
201	201/2016	Williamsville Drug Mart Limited	Kamie K. Ramkissoon
202	202/2016	The Medical Dispensary	George Dottin
203	203/2016	The Friendly Pharmacy Limited trading as St.	Lutchman Lalchan
		James Pharmacy	
204	204/2016	Kappa Drugs	Nicki De Bourg
205	205/2016	Sinanan's Drug Stop	Carlyle Sinanan
206	206/2016	Fabian's Pharmacy	Saeeda Ali
207	207/2016	Lisa's Pharmacy Limited	Jacqueline Jeffers-Ramjohn
208	208/2016	Medi Supplies Limited	Karen Teemal
209	209/2016	Chase Budget Drugs Limited	Surash Harripersad
210	210/2016	Hetero Drugs	Murali Mohan Reddy
			Gogireddy
211	211/2016	Health Mart Pharmacy	Sheryll Ann Balliramsingh
212	212/2016	Lays Drugs Limited	Alyssa Maria Ramadar
213	213/2016	Good Life Pharmacy	Carmel Travero
214	214/2016	A & R Budget Drugs	Cecil Mierez
215	215/2016	Croisee Pharmacy Limited	Rayavaram Siva Kumar Reddy
216	216/2016	Family Care Drug Mart Limited	Hassanali Soobrattie
217	217/2016	South Zone Pharmacy Limited	Ayoub Ali
218	218/2016	Macoya Pharmacy	Bernard Ammon
219	219/2016	Wellness Pharmacy Limited	Miriam Gordon
220	220/2016	Easipharm Limited	Joseann Mahabir
221	221/2016	Vernon Technologies Limited	Brenda Sylvester-Charles
222	222/2016	BPI Genethics Limited	Valini Ramcharan
223	223/2016	City Centre Pharmacy Limited	Teresina Lucio
224	224/2016	C & M Pharmacy Limited	Chandarbhan Maharaj
225	225/2016	Cindy's Drugs Limited	Lilawattee Jane Bhagwandin
226	226/2016	V & C Drug Mart Limited	Kishore Ramcharan
227	227/2016	Eddie's Pharmacy Limited	Emerson Ramjattan
228	228/2016	Customer Care Pharmacy	Bibi Rafeena Goolab-
			Rampersad
229	229/2016	Gulf View Pharmacy Limited	Satis Maharaj
230	230/2016	Bhagan's Drugs (1996) Limited	Marvin Chotai
231	231/2016	Valini's Drug Mart Limited	Valmiki Ramnarine
232	232/2016	Freeport Drugs Limited	Sharma Jaggernauth
233	233/2016	Hometown Pharmacy Limited	Yvonne Greene
234	234/2016	Bhagan's Drugs (Trincity) Limited	Jefferson Val Bishop
235	235/2016	Bhagan's Drugs (San Juan) Limited	Ved Ramlogan
236	236/2016	Health Site Pharmacy Company Limited	Vishnu Narine
237	237/2016	Palmyra Pharmacy Limited	Morisha Manick
238	238/2016	Lyn's Pharmacy Limited	Carolyn Agnoo Maharaj
239	239/2016	Palmyra Pharmacy Limited	Ranjit Seegobin
240	240/2016	Lyn's Pharmacy Limited	Khadija Dorset
241	241/2016	Med-Zone Pharmacy Limited	Annika Anushka Sankar

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
242	242/2016	Pills and Beyond Pharmacy Limited	Cherisse Mootoosingh
243	243/2016	Samodee's Drugstore	Shalini Mohanlal
244	244/2016	Mc Bean Pharmacy	Neela Boodoo
245	245/2016	Couva Pharm Limited	Cindy Heeralal
246	246/2016	Uptown Drugs Limited	Kamal Singh
247	247/2016	Young's Pharma-Serv Limited	Alan Young
248	248/2016	Young's Pharma-Serv Limited (Wholesale)	Alan Young
249	249/2016	Yogi Pharm	Yogitaa Ragoonath
250	250/2016	Eastern Pharmacy	Keshwar Maharaj
251	251/2016	Wellness Pharmacy Limited	Steve Ganessingh
252	252/2016	H & R Drugs	Subhash Bahadur Singh
253	253/2016	City Centre Pharmacy Limited	Enalin J. Aguinaldo-De Mesa
254	254/2016	P.S.I. Drugs	Indra Pamela Mohammed
255	255/2016	Samz Pharmafit Limited	Samantha Rajkumar
256	256/2016	Eddie's Pharmacy Limited	Deonarine Mayrhoo
257	257/2016	Southern Medical Services Company Limited	Debera Lackhan-Maraj
258	258/2016	SuperPharm Limited	Cherisse Ramsaran
259	259/2016	Apollo Pharmacy Limited	Raja Sekhara Reddy
			Sanikommu
260	260/2016	Total Care Pharmacy Limited	Ava Ramsaroop
261	261/2016	Gabriel's Pharmacy	Curtis Dann
262	262/2016	New Age Pharmacy	Avinash Garib
263	263/2016	Garcia's Drugs Limited	Felix Stephen Garcia
264	264/2016	Chaguanas Drug Mart Limited	Ken Baiju
265	265/2016	Diamond Vale Drugs Limited	Margarete Chang Kit
266	266/2016	Rampersad Medical Centre Limited	Jason Rampersad
267	267/2016	Prescription Plus Drugs Limited	Savera Ramsingh
268	268/2016	Pharmsav Drugs Limited	Nadia Bhaggan
269	269/2016	Live Well Pharmacy	Renuka Abdul
270	270/2016	Delta Drugs	Dale Sewlal
271	271/2016	Crichlow's Pharmacy	Carlhern Crichlow
272	272/2016	Express Drugs Limited	Jennifer Ragobar
273	273/2016	Healing Garden Pharmacy	Anita Ramkelawan
274	274/2016	City Centre Pharmacy Limited	Gail Ramnath
275	275/2016	Value Plus Pharmacy Limited	Cristal Sooknanan
276	276/2016	Horace's Drugs	Horace Ramjattan
277	277/2016	Tunapuna Pharmacy and Cosmetic Centre	Brent Roopchandsingh
		Limited	
278	278/2016	Eddie's Pharmacy Limited	Vijay Soodeen
279	279/2016	Roystonia Pharmacy	Renuka Mahadeo
280	280/2016	Pharmtech Caribbean Limited	Caroline Maharaj
281	281/2016	Edinburgh Pharmacy Limited	Ravi Sobie
282	282/2016	Specialist Pharmacy Limited	Indira Singh-Bhagaloo
283	283/2016	Police Health Facility	Marcelle Charles

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
284	284/2016	Green Leaf Pharmacy Limited	Maida Lalic-Samaroo
285	285/2016	R & RS Drugs	Rahzia Sieuchand
286	286/2016	Living Water Community Hospice	Richard Anthony Clerk
			(Medical Doctor)
287	287/2016	Phamacare Drugs Limited	Jairam Maharaj
288	288/2016	Rite-Aid Pharmacy Limited	Dennis Ramkoon
289	289/2016	Quality Care Pharmacy	Narindra Maharaj
290	290/2016	Williams Drug Store Limited	Trevor Williams
291	291/2016	V & H Pharmacy	Gyandeo Shukla
292	292/2016	My Medicine Bag Limited	Natasha Nanan-Rampersad
293	293/2016	Pharma Rx Pharmacy	Mathias Marfan
294	294/2016	Felicity Community Pharmacy	Nigel Harnarine
295	295/2016	Couva Pharm Limited	Shazim Tullah
296	296/2016	Cunupia Drug Stop	Gobin Gopaul
297	297/2016	Balmain Drug Mart	Curtis Victor Young Lum Kin
298	298/2016	Ali's Pharmacy Limited	Shaun Vincent
299	299/2016	Ali's Pharmacy Limited	Michael Allum
300	300/2016	Ali's Pharmacy Limited	Meela Ali
301	301/2016	Couva Drug Mart Limited	Ria Dhanoolal
302	302/2016	Reliant Pharmacy	Ronald Tsoi-A-Fatt
303	303/2016	Tropical Pharmacy Limited	Ricky Roodal
304	304/2016	Seeta's Pharmacy	Dhanrajie Ragoonanan
305	305/2016	Busy Corner Drugs	Steve Ramlogan
306	306/2016	Pharmacy Plus	Kerra Dindial
307	307/2016	Water Sprout Pharmacy & Wellness Center	Denise Mohammed
308	308/2016	R & J Drugs Limited	Judy Naraynsingh
309	309/2016	City Drugs Investments Limited	Wilson D. Reyes
310	310/2016	D Village Pharm Limited	Rhea Ragobar
311	311/2016	Lee's Pharmacy Limited	Miguel Bujhawan
312	312/2016	Lopinot Drugs Limited	Dineash Bidaisee
313	313/2016	P.C.L. Pharmacy	Glenwayne Suchit
314	314/2016	Pharmacy Connection Limited	Navin Deonarine
315	315/2016	Quik-Pharm	Alana Bosland
316	316/2016	Universal Pharmacy	Jimmy B. Cosico
317	317/2016	The Royal Pharmacy Limited	Allan Hewitt
318	318/2016	Whole Life Pharmacy Limited	Alwyn Noble
319	319/2016	Glendale's Drugs	Glen Sewlal
320	320/2016	Lisa's Drug Mart	Gale Ahiliya Ramsumair
321	321/2016	MWL Incorporated Limited trading as The	Dinesh Moonan
		Drug Store, Rio Claro	
322	322/2016	Charlotte Drugs Limited	Addison Jagessar
323	323/2016	Unique Pharmacy	Varma Boodhai
324	324/2016	Ferrisa's Pharmacy	Fayad Mohammed
325	325/2016	Westmed Pharmacy	Mohan Kelvis Bentinck

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
326	326/2016	Medical Associates Central Hospital Limited	Hemant Kumar Persad
		-	(Medical Doctor)
327	327/2016	Mega Drugs Limited	Laila Seecharan
328	328/2016	Vark Enterprises Limited trading as Medicine	Anil Bheem
		Plus	
329	329/2016	Tacarigua Pharmacy West	Terrence Gooding
330	330/2016	Capital Drugs Limited	Pradeep Mallireddy
331	331/2016	Bournes Road Pharmacy	Andrew Rahaman
332	332/2016	St. Joseph Pharmacy	Shyam Ramoutar
333	333/2016	Waterfront Pharmacy	Auldith Beache
334	334/2016	St. Helena Drug Mart	Nalini Dial
335	335/2016	San Juan Pharmacy Limited	Khemraj Vishal Lakhan
336	336/2016	Kappa Drugs	Andrew Ramirez
337	337/2016	Fit For Life Pharmacy Limited	Robert Soogrim
338	338/2016	Med-X Drugs Limited	Winstondave O'Nardex
339	339/2016	Tacarigua Pharmacy	Gokale Valmiki Dial
340	340/2016	Pharmacy Plus	Anusha Beharrysingh
341	341/2016	Med-Farm Drugs Limited	Juliet M. Marbella
342	342/2016	Complete Care Pharmacy Services	Roger Cain
343	343/2016	O'Meara Pharmacy Limited	Anne-Marie Doyle
344	344/2016	Twin Island Pharmacy Limited	Amelia Birusingh-Cipriani
345	345/2016	Nancy's Pharmacy Limited	Natasha Singh Ramsook
346	346/2016	Verison Drugs Limited	Patrick Harris
347	347/2016	Med-Farm Drugs Limited	Nellie Singh-Musai
348	348/2016	Drugs & Beyond Limited	Savitri Sandra Ramnanan
349	349/2016	Ali's Pharmacy Limited	Prunella Watson-Woods
350	350/2016	Ali's Pharmacy Limited	Maurice Douglas
351	351/2016	Fyzabad Pharmacy Limited	Crystal Tahal
352	352/2016	Tablet City Limited	Carol Elizabeth Ramsoomair
353	353/2016	Penny Med	Ambika Christine Sooker
354	354/2016	Saddle Pharmacy	Kareem Wesley Diaz
355	355/2016	The Medicine Stop	Judy Dass-Ali
356	356/2016	Medac Drugs Limited	Margaret Edghill
357	357/2016	Ramkoon's Pharmacy	Desmond Ramkoon
358	358/2016	Khan's Pharmacy	Ameena Ali
359	359/2016	Hatha Health Pharmacy Limited	Bennett Kangalee
360	360/2016	Scarborough General Hospital	Beverly Dass-Vilain
361	361/2016	Victoria Nursing Home Limited	Dr. Clem Calvin Ragobar
			(Medical Practitioner)
362	362/2016	Petrotrin Augustus Long Hospital Dispensary	Rhonda Bharath
363	363/2016	Petrotrin Medical Centre Dispensary Pointe-	Ilva Crooks
		A-Pierre	
364	364/2016	Petrotrin Trinmar Operations Medical Centre	Elizabeth Huggins-Andrews
365	365/2016	Petrotrin Medical Stores	Diane Julien

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
366	366/2016	Petrotrin Medical Centre Dispensary	Bertina Hernandez
		Guayaguayare	
367	367/2016	Lisa's Pharmacy	Kafton Scott Ramsamooj
368	368/2016	The Surgi-Med Clinic	Dr. Anirudh Mahabir
			(Ophthalmologist)
369	369/2016	Anva's Health Plus Pharmacy	Vandita Bissoonlal-Dubrey
370	370/2016	Princes Town Drug Mart Limited	Susan Khan
371	371/2016	Health Plus Pharmacy Limited	Shobha Rampersadsingh
372	372/2016	Medicine Corner Limited	Neil Varma Persad
373	373/2016	Cama Enterprises Limited trading as	Rosanne Nanton
		Medicine Box	
374	374/2016	Prescription House Limited	Sunita Hardit-Singh
375	375/2016	You and Us Pharmacy Limited	Marlon Austin
376	376/2016	Petrotrin Medical Centre Dispensary Forest	Kenlyn Hazel Keppel
		Reserve	
377	377/2016	Petrotrin Medical Centre Dispensary Point	Selwyn Carr
		Fortin	
378	378/2016	Petrotrin Medical Centre Dispensary Penal	Christine Corbie
379	379/2016	Petrotrin Medical Centre Dispensary Brighton	Patricia Elvin
380	380/2016	Petrotrin Medical Centre Dispensary Santa	Jacqueline Ifill
		Flora	
381	381/2016	Shalisa Pharmacy Limited	Sreenivasa Reddy Golamari
382	382/2016	V & S Pharmaceuticals Limited	Varune Persad
383	383/2016	NIPDEC – Central Stores	Colin Clifton
384	384/2016	Hyper Pharm Limited	Andrew Sam
385	385/2016	A & L Pharmacy (2005) Limited	Clyde Albert Allum
386	386/2016	Oscar Francois Limited	Brent Scott
387	387/2016	Freeport Medical Centre Limited	Dr Rohit Dass (Medical
			Doctor)
388	388/2016	Trinidad And Tobago Forensic Science	Arlette Lewis (Director)
		Centre	
389	389/2016	Pharm-Aid Company Limited trading as	Rhonda Karim
		Apex Pharmacy	
390	390/2016	Oscar Francois Limited	Kenny Arrindell
391	391/2016	Livingston Drugs Limited	Joslin Brewster
392	392/2016	La Romain Drug Mart Limited	Nirupa Babwah
393	393/2016	AP Scott Trinidad Limited	Amit Bhalekar
394	394/2016	M & R Pharmacy	Meera Heeraman
395	395/2016	A & N Drug Mart Limited	Anilla Ananda Ballyram
396	396/2016	D. Bartlett Drugs Limited	Daphne Bartlett
397	397/2016	Lisa's Pharmacy	Kavita Sewnath
398	398/2016	Alphamed Diagnostic Services Limited	Raja Siva Prasad Muvvalla
			(Medical Doctor)
399	399/2016	Pharmacy & Beyond Limited	Vedwattee Ramsaran

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
400	400/2016	Zion Drug Mart	Kiash Harryram
401	401/2016	Park View Drugs	Ravi Ramanand
402	402/2016	Southside Pharmacy	Llana Philip
403	403/2016	We Care Pharmacy Limited	Rai Ragoonanan
404	404/2016	NIPDEC – Central Stores	Jayakrishnan Sreekumar
405	405/2016	Easipharm Limited	Pamela Townsend
406	406/2016	Roger's Drug Store Limited	Curron Lum Kin
407	407/2016	Health Image Pharmacy	Sarvy Ramlogan
408	408/2016	Family Care Pharmacy	Ernest Sankar
409	409/2016	Health Site Pharmacy Company Limited	Jerard Castillo
410	410/2016	A+ Drugs Limited	Rajiv Rudra Bhagwat
411	411/2016	SuperPharm Limited	Sharon Thomson
412	412/2016	Bryden pi Limited	Carin Flores
413	413/2016	City View Pharmacy Limited	Nisha Khan-Roopnarine
414	414/2016	Kappa Drugs	Samantha Bally
415	415/2016	Karimed Pharmaceuticals Limited	Claire Crevier
416	416/2016	San Fernando General Hospital	Roy Umraw
417	417/2016	Eric Williams Medical Sciences Complex	Michelle Ramsay-James
418	418/2016	Jevestus Drugs	Claudette Browne
419	419/2016	Marathon Health	Akash Rambharose
420	420/2016	CVA Pharmacy	Janelle Reid
421	421/2016	A.D.C. Pharmacy	Rahaana Rajack
422	422/2016	Muhammad's Pharmacy	Shalima Mohammed
423	423/2016	Felicity Community Pharmacy	Meera Singh
424	424/2016	Bhagan's Drugs (Tobago) Limited	Raissa Hassiman
425	425/2016	Alive Pharmacy	Ravi Nanhu
426	426/2016	O'Meara Pharmacy Limited	Adona Boodhai
427	427/2016	Easipharm Limited	Cuthbert Ferdinand
428	428/2016	NIPDEC – Central Stores	Marissa Austin
429	429/2016	Pharmdynamics Drugs (PD Drugs) Limited	Jimmy B. Cosico
430	430/2016	Massy Stores Pharmacy	Susan Jones-Lashley
431	431/2016	Shalisa Pharmacy Limited	Karyn Indira Mohan-Bridglal
432	432/2016	Williams Drug Store Limited	Anne Marie Walters
433	433/2016	Drugs & Beyond Limited	Hema Ramcharan
434	434/2016	Baggies Pharmacy Limited	Harold K. Bhagwandass
435	435/2016	Ellie's Medicine Pharmacy Limited	Nigel Harnarine
436	436/2016	Healing Garden Pharmacy Limited	Anita Ramkelawan
437	437/2016	Tara's Pharmacy Limited	Kavita Ragoobar
438	438/2016	Vitadrugs Limited	Neil Ahwai
439	439/2016	Medwin Pharmacy	Anil Kumar Reddy Vaka
440	440/2016	Ocean Rose Pharmacy Limited	Leyya S. Ali
441	441/2016	Couva Drug Mart Limited	Carol Elizabeth Ramsoomair
442	442/2016	R & R Budget Drugs Limited	Nadia Deonarine
443	443/2016	Valsayn Pharmacy Limited	Salina Dhanoolal

NO.	S&D#	NAME OF ORGANISATION	NAME OF LICENSEE
444	444/2016	Ava's Total Care Pharmacy Limited	Anushika L. Sooker
445	445/2016	Bhagan's Drugs (Price Plaza) Limited	Shiva Kadaroo
446	446/2016	Capital Drugs Limited	Ryan Rajiv Ramsahai
447	447/2016	CAMA Enterprises Limited trading as	Joanne Salandy
		Medicine Box	
448	448/2016	Wilmed Limited	Pamela Townsend
449	449/2016	Pharmakon Drugs Limited	Videsh Maharaj
450	450/2016	H & R Drugs	Jameel Rahaman
451	451/2016	Lyn's Pharmacy Limited	Khadija Dorset
452	452/2016	The Family Medicine Centre Limited	Taras Vishal Lalchan